Author’s response to reviews

Title: Measuring the health impact of human rights violations related to Australian asylum policies and practices: a mixed methods study

Authors:

Vanessa Johnston (vanessa.johnston@menzies.edu.au)
Pascale Allotey (pascale.allotey@brunel.ac.uk)
Kim Mulholland (Kim.Mulholland@lshtm.ac.uk)
Milica Markovic (milica.markovic@med.monash.edu.au)

Version: 3 Date: 14 January 2009

Author’s response to reviews:

Editor
BMC International Health and Human Rights

14th January, 2009

Dear Editor,

Please find attached the revised manuscript ‘Measuring the health impact of human rights violations related to Australian asylum policies: a mixed methods study’ (Version 2) for consideration.

The revisions made to this paper are presented in tracked changes for ease of recognition. We have complied with all of the suggested changes made by the reviewer (Edward Mills):

1. We have included all of the exact p-values in the relevant tables and have deleted asterisks indicating threshold p-values.

2. We have deleted page numbers in the main text which relate to quotations.

3. We have deleted the quote from Mann on p.21.

We look forward to hearing from you about the outcome of this paper.

Yours sincerely,

Dr. Milica Markovic
Submitting author
Senior Research Fellow
School of Psychology, Psychiatry and Psychological Medicine